Title: Acute onset of intracranial subdural hemorrhage five days after spinal anesthesia for knee arthroscopic surgery: a case report

Authors:

Tetsuo Hagino (tmhagino@amber.plala.or.jp)
Satoshi Ochiai (hxcmk230@ybb.ne.jp)
Yoshiyuki Watanabe (yw01d08@yahoo.co.jp)
Shinya Senga (shinyasenga0721@yahoo.co.jp)
Masanori Saito (msaitoh@yamanashi.ac.jp)
Hirofumi Naganuma (naganuma-h@nhokoufu.hosp.go.jp)

Version: 2 Date: 29 July 2011

Author's response to reviews:

Jury 29, 2011

Journal of Medical Case Report,

Dear Editor:

My co-author and I submit the enclosed manuscript, Acute onset of intracranial subdural hemorrhage five days after spinal anesthesia for knee arthroscopic surgery: a case report, for publication in Journal of Medical Case Report.

Intracranial subdural hemorrhage after spinal anesthesia occur rarely. Moreover, pattern with acute onset and no early stage headache was extremely rare, and there is no detailed report. We present a case of acute onset of intracranial subdural hematoma without trauma five days after spinal anesthesia.

We certify that this article is original, that it is not under consideration by another journal, and that it has not been previously published, and we agree to transfer copyright to Journal of Medical Case Report, in the event that this manuscript is accepted for publication.

Thank you for your consideration of our paper. We look forward to hearing from you.

Sincerely,

Tetsuo Hagino, MD, PhD